Round 1

Dear reviewer:

Great thanks for your appreciation and suggestions for our manuscript. Our response for the comments is as follows.

1. **Suggestions 1.** Considering the oxidative stress and inflammation play the critical roles in the obesity and insulin resistance (i.e.: metabolic syndrome)-mediated additional pathogenesis to T1D, a recent study (PMID: 37298384) to show the MetS as a factor of impairment of antioxidant defense system in youth with T1D may be included into this editorial with a short discussion as the one mechanism why Adiposity & IR will exacerbate pathogenesis in T1D.

   **Answers:** Great thanks for your suggestions, we have added a short discussion in the introduction section regarding the role of oxidative stress and inflammation in adiposity-mediated pathogenesis in T1D complications. The content is highlighted in the text.

2. **Suggestions 2.** The reviewer suggests addition of one or two figure/table to outline the editorial contents, which will help the readers’ understanding and impression.

   **Answers:** Thanks very much for your comments. We agree that figures and tables facilitate understanding of the audience in reviews and articles. Nevertheless, the purpose of this editorial is primarily commenting on a published research article in the World Journal of Diabetes. The author team consider it may not be necessary in this particular case. We are more than happy to illustrate the idea via figures and tables in a full-length paper, either narrative review or original studies.

Cheers

Sheyu Li
Round 2

Although the authors took my first suggestion by adding one sentence with citation, and did not take my second suggestion with a very weak reason, I remain accept it. My final review conclusion is "acceptable"

Dear Reviewer:

Great thanks for your suggestions, we have added a figure outlined the content of the outline the editorial contents to help the readers’ understanding and impression.

Cheers
Sheyu Li